Author’s response to reviews

Title: Identification of gene mutations in patients with primary periodic paralysis using targeted next-generation sequencing

Authors:
Sushan Luo (susanro_36@hotmail.com)
Minjie Xu (11210700016@fudan.edu.cn)
Jian Sun (sunjian19927@yeah.net)
Kai Qiao (kateqiao@hotmail.com)
Jie Song (sdsj1990@163.com)
Shuang Cai (shuang4601441@126.com)
Wenhua Zhu (jessiblue@yeah.net)
Lei Zhou (zhoulei0156022@hotmail.com)
Jianying Xi (xijiangying@fudan.edu.cn)
Jiahong Lu (lujiahong@fudan.edu.cn)
Xiaohua Ni (nixiaohua@sippr.org.cn)
Tonghai Dou (tonghaidou@fudan.edu.cn)
Chongbo Zhao (zhao_chongbo@fudan.edu.cn)

Version: 1 Date: 17 Jul 2018

Author’s response to reviews:

Dear Editors,

Many thanks for your detailed editing of the original manuscript and good suggestions.

We modified the manuscript and cover letter accordingly. Please find the revised version and do not hesitate to contact us if we need further modifications.

Best!
Chongbo Zhao